Author's response to reviews

Title: Gene-enhanced tissue engineering for dental hard tissue regeneration: (1) Overview and practical considerations

Authors:

Paul C. Edwards (pedwards@creighton.edu)
James Mason (anjmason@optonline.net)

Version: 2 Date: 10 May 2006

Author's response to reviews:

Dear Prof. Meyer:

We wish to thank you and Prof. Wiesmann for your thorough review of our submitted manuscripts. The recommended revisions have been included in our revised submissions.

Sincerely,

Paul C. Edwards